CS 282A/MATH 209A: Foundations of Cryptography Prof. Rafail Ostrovsky

Handout 1: Student Survey

10 January 2005 prepared by Zaphod Beeblebrox

Please fill in all the information. Please write legibly.

1. Name: 2. Year:

3. Deparment/Major: 4. Email Address: 5. Phone:

6. Taking this class for credit? (circle one:) yes no maybe 7. Are you familiar with NP-reductions? (circle one:) yes no 8. Have you taken CS180? (circle one:) yes no 9. If not, why do you think that you are qualified to take this class?

10. Have you had any other classes you believe to be relevant?

11. Name an NP-complete problem (or write “don’t know”): 12. Why are you taking this class?

13. Are you satisfying a requirement with this class? (circle one:) yes no If so, what grade do you need? (Note: No promises....)

1-1